29 July 2021

Editor in Chief
World Journal of Hepatology

Dear Sir,

We are pleased to have been given the chance to revise our manuscript No. 68940, entitled “Rare primary mature teratoma of the liver: A case report” for publication in World Journal of Hepatology.

We also appreciate the constructive comments from the reviewer.

We addressed the reviewer’ comments and revised the manuscript, accordingly, based on the recommendations and suggestions.

A response to the reviewer’ comments is provided below.

We hope that the revised version, which includes tables and figures, will meet the requirements for publication.

Sincerely yours,

Yury O Zharikov, MD, PhD, MBA
Associate professor, Department of Human Anatomy,
I.M. Sechenov First Moscow State Medical University (Sechenov University),
8-2 Trubetskaya Street,
Moscow, Russian Federation, 119435
TEL: +7 -999-912—11-45
Email: dr_zharikov@mail.ru
Reviewer #1:

**Scientific Quality:** Grade B (Very good)

**Language Quality:** Grade A (Priority publishing)

**Conclusion:** Accept (General priority)

**Specific Comments to Authors:** Thanks for the paper, is very well written. I would recommend only to correct the paragraphs layout in the case presentation part (ex. after “history of past illness” insert a “:” and do not go down with spacing.

Thank the reviewer for this important comment.

Regarding your suggestion, the system Automatically Generate Full Text Files corrected this point in the text of the manuscript.